Author's response to reviews

Title: User needs elicitation via analytic hierarchy process (AHP). A case study on a CT scanner.

Authors:

Leandro Pecchia (leandro.pecchia@nottingham.ac.uk)
Jennifer Martin (jennifer.martin@nottingham.ac.uk)
Angela Ragozzino (angelarag56@virgilio.it)
Carmele Vanzanella (carmela.vanzanella@cnr.it)
Luciano Mirarchi (luciano.mirarchi@siemens.com)
Arturo Scognamiglio (Mario.Scognamiglio@rummo.it)
Steve Morgan (steve.morgan@nottingham.ac.uk)

Version: 3 Date: 6 August 2012

Author's response to reviews: see over
Dear Editor,

Please find attached the paper entitled “User needs elicitation via analytic hierarchy process (AHP). A case study on a CT scanner”, which are resubmitting for publication in BMC Medical Informatics and Decision Making, after carefully considering reviewers comments.

We would like to thank you and the two reviewers very much for their detailed reviews of our previous submission. These have been extremely helpful in revising and improving the paper. Please find attached a document containing a list of the suggestions made by the reviewers to and our response to this. In addition to the changes suggested by the reviewers, we have taken this opportunity to tighten up the paper as whole, and we feel the paper is much better as a result. All the changes made are highlighted in yellow in the manuscript re-submitted.

We hope that we have adopted the suggestions made by the reviewers satisfactorily, but please do not hesitate to contact us if you require any further information. We look forward to hearing from you and we would like to thank you for your consideration of the revised manuscript.

Sincerely,

Leandro Pecchia

(on behalf of all the authors)